Author's response to reviews

Title: A Survey of Variable Selection Methods in Two Chinese Epidemiology Journals

Authors:

Huimin Liao (082102094@fudan.edu.cn)
Henry S Lynn (hslynn@shmu.edu.cn)

Version: 3 Date: 22 September 2010

Author's response to reviews:

Formatting changes to be made:

-------------------------------

Major revisions (we require the author to make these changes)
Conclusion: Please include a 'Conclusions' section after the Discussion.
The last paragraph of the Discussion has now be made into the Conclusions paragraph.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)
Tracking: Please remove this from the manuscript.

Done.